

# G OPEN ACCESS

**Citation:** Bhatia A, Preiss AJ, Xiao X, Brannock MD, Alexander GC, Chew RF, et al. (2025) Effect of nirmatrelvir/ritonavir (Paxlovid) on hospitalization among adults with COVID-19: An electronic health record-based target trial emulation from N3C. PLoS Med 22(1): e1004493. https://doi.org/ 10.1371/journal.pmed.1004493

Academic Editor: Aaloke Mody, Washington University School of Medicine, UNITED STATES OF AMERICA

Received: February 20, 2024

Accepted: October 21, 2024

Published: January 17, 2025

**Copyright:** © 2025 Bhatia et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All data is available in the N3C Data Enclave to those with an approved protocol and data use request from an institutional review board. Data access is governed under the authority of the National Institutes of Health; more information on accessing the data can be found at https://covid.cd2h.org/for-researchers. See Haendel et. al. for additional detail on how data is ingested, managed, and protected within the N3C Data Enclave (DOI: 10.1093/jamia/ocaa196). Code RESEARCH ARTICLE

Effect of nirmatrelvir/ritonavir (Paxlovid) on hospitalization among adults with COVID-19: An electronic health record-based target trial emulation from N3C

Abhishek Bhatia<sup>1</sup><sup>•</sup>\*, Alexander J. Preiss<sup>2</sup>, Xuya Xiao<sup>3</sup>, M. Daniel Brannock<sup>2</sup>, G. Caleb Alexander<sup>3</sup>, Robert F. Chew<sup>2</sup>, Hannah Davis<sup>4</sup>, Megan Fitzgerald<sup>4</sup>, Elaine Hill<sup>5</sup>, Elizabeth P. Kelly<sup>1</sup>, Hemalkumar B. Mehta<sup>3</sup>, Charisse Madlock-Brown<sup>6</sup>, Kenneth J. Wilkins<sup>7,8</sup>, Christopher G. Chute<sup>3</sup>, Melissa Haendel<sup>1</sup>, Richard Moffitt<sup>9</sup>, Emily R. Pfaff<sup>1</sup>, the N3C Consortium<sup>1</sup>

1 University of North Carolina at Chapel Hill, Chapel Hill, North Carolina, United States of America, 2 RTI International, Durham, North Carolina, United States of America, 3 School of Medicine, Public Health, and Nursing, Johns Hopkins University, Baltimore, Maryland, United States of America, 4 Patient-Led Research Collaborative, New York, New York State, United States of America, 5 University of Rochester, Department of Public Health Sciences and Department of Economics, Rochester, New York State, United States of America, 6 University of Tennessee Health Science Center, Memphis, Tennessee, United States of America, 7 National Institute of Diabetes & Digestive & Kidney Diseases, Office of the Director, National Institutes of Health, Bethesda, Maryland, United States of America, 8 F. Edward Hébert School of Medicine, Department of Preventive Medicine & Biostatistics, Uniformed Services University of the Health Sciences, Bethesda, Maryland, United States of America, 9 Emory University, Atlanta, Georgia, United States of America

• These authors contributed equally to this work.

¶ Membership of the the N3C Consortium is listed in the Acknowledgments. \* abhatia@unc.edu

# Abstract

# Background

Nirmatrelvir with ritonavir (Paxlovid) is indicated for patients with Coronavirus Disease 2019 (COVID-19) who are at risk for progression to severe disease due to the presence of one or more risk factors. Millions of treatment courses have been prescribed in the United States alone. Paxlovid was highly effective at preventing hospitalization and death in clinical trials. Several studies have found a protective association in real-world data, but they variously used less recent study periods, correlational methods, and small, local cohorts. Their estimates also varied widely. The real-world effectiveness of Paxlovid remains uncertain, and it is unknown whether its effect is homogeneous across demographic strata. This study leverages electronic health record data in the National COVID Cohort Collaborative's (N3C) repository to investigate disparities in Paxlovid treatment and to emulate a target trial assessing its effectiveness in reducing severe COVID-19 outcomes.

# Methods and findings

This target trial emulation used a cohort of 703,647 patients with COVID-19 seen at 34 clinical sites across the United States between April 1, 2022 and August 28, 2023. Treatment

Availability The N3C Enclave is available for public use. To access data used within this manuscript, institutions must have a signed Data Use Agreement executed with the U.S. National Center for Advancing Translational Sciences (NCATS) and their investigators must complete mandatory training and must submit a Data Use Request (DUR) to N3C. To request N3C data access, researchers must follow instructions at https:// covid.cd2h.org/onboarding. Code is available to those with valid login credentials for the N3C Data Enclave. It was written for use in the enclave on the Palantir Foundry platform (https://www.palantir. com/platforms/foundry/), where the analysis can be reproduced by researchers. It can be exported for review upon request.

**Funding:** National Center for Advancing Translational Sciences, Award Number: 3U24TR002306-04S2; Recipient: M.H. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

Abbreviations: ARD, absolute risk difference; CCI, Charlson comorbidity index; CI, confidence interval; COVID-19, Coronavirus Disease 2019; CWBI, community well-being index; EHR, electronic health record; EUA, Emergency Use Authorization; FDA, Food and Drug Administration; HR, hazard ratio; IPCW, inverse probability of censoring weights; mAb, monoclonal antibodies; N3C, National COVID Cohort Collaborative; NNT, number needed to treat; NIH, National Institute of Health; RR, relative risk; RRR, relative risk reduction; SARS-CoV-2, Severe Acute Respiratory Syndrome Coronavirus 2; TTE, target trial emulation; VA, Veterans Affairs. was defined as receipt of a Paxlovid prescription within 5 days of the patient's COVID-19 index date (positive test or diagnosis). To emulate randomization, we used the clone-censor-weight technique with inverse probability of censoring weights to balance a set of covariates including sex, age, race and ethnicity, comorbidities, community well-being index. (CWBI), prior healthcare utilization, month of COVID-19 index, and site of care provision. The primary outcome was hospitalization; death was a secondary outcome. We estimated that Paxlovid reduced the risk of hospitalization by 39% (95% confidence interval (CI) [36%, 41%]; p < 0.001, with an absolute risk reduction of 0.9 percentage points (95% CI [0.9, 1.0]; p < 0.001), and reduced the risk of death by 61% (95% CI [55%, 67%]; p < 0.001), with an absolute risk reduction of 0.2 percentage points (95% CI [0.1, 0.2]; p < 0.001). We also conducted stratified analyses by vaccination status and age group. Absolute risk reduction for hospitalization was similar among patients that were vaccinated and unvaccinate, but was much greater among patients aged 65+ years than among younger patients. We observed disparities in Paxlovid treatment, with lower rates among black and Hispanic or Latino patients, and within socially vulnerable communities. This study's main limitation is that it estimates causal effects using observational data and could be biased by unmeasured confounding.

# Conclusions

In this study of Paxlovid's real-world effectiveness, we observed that Paxlovid is effective at preventing hospitalization and death, including among vaccinated patients, and particularly among older patients. This remains true in the era of Severe Acute Respiratory Syndrome Coronavirus 2 (SARS-CoV-2) Omicron subvariants. However, disparities in Paxlovid treatment rates imply that the benefit of Paxlovid's effectiveness is not equitably distributed.

# Author summary

# Why was this study done?

- Paxlovid is widely used to prevent severe Coronavirus Disease 2019 (COVID-19), but its real-world effectiveness is uncertain.
- Prior studies differed in their findings, and none of them used a large, nationally sampled, and recent cohort.
- It is unknown whether Paxlovid may be more or less effective in certain demographic groups.

# What did the researchers do and find?

• We used a cohort of 703,647 patients from the National Covid Cohort Collaborative's (N3C's) electronic health record database to estimate Paxlovid's real-world effectiveness and investigate disparities in treatment.

- We used the target trial emulation technique to estimate the causal effect of Paxlovid treatment using observational data.
- We found that Paxlovid treatment reduced the risk of hospitalization by 39% and the risk of death by 61%.
- We found that Paxlovid reduced the risk **of hospitalization** by a similar **absolute** amount regardless of whether the patient was vaccinated prior to getting COVID-19.
- We found that Paxlovid reduced risk more among patients aged 65+ years than among younger patients.
- We found disparities in Paxlovid treatment, with lower rates among black and Hispanic or Latino patients, and within socially vulnerable communities.

# What do these findings mean?

- Paxlovid is effective at preventing severe COVID-19 and is effective against current variants.
- Fully vaccinated people may have a further reduced risk of severe outcomes by receiving Paxlovid.
- People aged 65 years or older may benefit the most from Paxlovid treatment.
- However, disparities in Paxlovid treatment rates imply that the benefit of Paxlovid's effectiveness is not equitably distributed.
- This study's main limitation is that it estimates causal effects, but it is not randomized like a clinical trial. Instead, we control for other variables which could bias the estimate, but if we missed some important variables, the estimates could be incorrect.

### Introduction

The Coronavirus Disease 2019 (COVID-19) pandemic has had a profound global impact, with over 775 million cases and 7 million deaths as of 24 April 2024 [1]. This crisis has been met with research and drug development efforts at unprecedented speed, resulting in a number of new treatments aimed at lessening the risk of progression to severe disease. One such treatment, Paxlovid, is a combination of nirmatrelvir, an inhibitor of the main Severe Acute Respiratory Syndrome Coronavirus 2 (SARS-CoV-2) protease, and ritonavir, a pharmacological booster. In December 2021, the United States Food and Drug Administration (FDA) issued an Emergency Use Authorization (EUA) for Paxlovid, enabling its prescription to high-risk SARS-CoV-2–positive patients aged 12 years and older [2]. The EUA was based on the phase II-III EPIC-HR (Evaluation of Protease Inhibition for Covid-19 in High-Risk Patients) trial, which reported an 88.9% reduction in the risk of COVID-related hospitalization or death among those who received Paxlovid compared to those who received placebo [3].

Since the EUA issuance, several studies have assessed the effectiveness of Paxlovid using real-world data. An electronic health record (EHR)-based study in the Kaiser Permanente

Southern California health system found that fewer than 1% of patients treated with Paxlovid (n = 5,287) were hospitalized or seen in the emergency department within 5 to 15 days of the drug being dispensed, though this was not compared with an untreated group [4]. An early retrospective cohort study in Hong Kong also found that patients treated with Paxlovid (n =4,921) were at decreased risk of hospitalization, albeit with a much smaller risk reduction of 21% [5]. Another early study leveraging a large repository of Israeli health care data found a hazard ratio (HR) of 0.54 for severe COVID outcomes in patients treated with Paxlovid (n =4,737) when compared with untreated patients, showing a protective association, but at a lower magnitude than the original EPIC-HR analysis [6]. A similar Israeli study, disaggregated by age, found a weaker protective association between Paxlovid and hospitalization within a cohort of patients aged 40 to 64 years (n = 66,433, HR = 0.74), and a stronger association for patients 65 years and over (n = 42,821, HR = 0.27) [7]. More recently, a study of 68,876 Cleveland Clinic patients with COVID-19 diagnoses from April 2022 to February 2023 found an adjusted hazard ratio of hospitalization of 0.63. That study was widely interpreted in news media as suggesting that Paxlovid may be less effective at preventing hospitalization caused by SARS-CoV-2 Omicron subvariants, although several earlier studies had similar findings [8,9]. To our knowledge, 2 studies have explicitly estimated Paxlovid's causal effect on hospitalization in the real world. The first used a cohort of 44,551 patients from the Mass General Brigham health system in the northeastern US [10]. The second used a cohort of 256,288 patients from the US Department of Veterans Affairs (VA) database [11]. Both studies found a risk ratio of 0.60 for hospitalization. However, their study periods included the Omicron wave and ended in 2022, so the findings may not apply to current variants. The veteran population also differs from the general population in several relevant ways, including frequent history of traumatic brain injury, environmental exposures, and differing healthcare utilization patterns [12-16]. Together, these studies show a clear association between Paxlovid and reduced COVID-19 severity in real-world settings, but there remains a dearth of research specifically aimed at understanding the treatment effect of Paxlovid on COVID-19 outcomes with large, contemporary, national samples derived from real-world data.

Several EHR studies have also used real-world data to uncover racial and social disparities in the prescription of various COVID-19 treatments in the United States, including Paxlovid. Prior to Paxlovid's authorization, multiple studies noted racial and social disparities among SARS-CoV-2-positive patients in access to and treatment with monoclonal antibodies (mAb), with black and Hispanic or Latino patients less likely than white patients to receive treatment with mAb [17–19]. A more recent large-scale study of EHR data revealed that those disparities have persisted with Paxlovid; from April through July of 2022, the rate of Paxlovid treatment was 35.8% lower among black adult patients than white adult patients [20].

Through the National Institute of Health's (NIH) National COVID Cohort Collaborative (N3C), we expanded on prior studies of Paxlovid effectiveness and treatment patterns. We used the target trial emulation (TTE) framework and a large geographically and demographically diverse cohort from N3C's EHR data repository [21]. Our study period of April 2022 through August 2023 provides up-to-date information to better assess Paxlovid's effectiveness in the era of Omicron subvariant dominance. This study consisted of several analyses. First, we characterized the population prescribed Paxlovid and assessed potential disparities in Paxlovid prescription. Second, we estimated the causal effect of Paxlovid treatment on hospitalization and death among adults with COVID-19 in the US [21]. Third, building on prior work, we conducted a stratified analyses to estimate Paxlovid's effect across vaccination status and age groups, that, to our knowledge, addresses an important gap in understanding the heterogeneity of its effectiveness across diverse patient populations.

### Methods

We performed a target trial emulation to assess the effect of Paxlovid treatment within 5 days of COVID-19 index on the risk of severe COVID-19 outcomes within 28 days of COVID-19 index. We followed a two-step process for emulating target trials with observational data: first, we articulated the causal question of interest in the form of a hypothetical randomized trial protocol, specifying eligibility criteria, treatment strategies, treatment assignment, the study period for follow-up, the outcome of interest, causal contrasts, and the analysis plan to estimate effects (Table 1) [22]. Second, we emulated each component of this protocol using patient-level data inside the NIH-hosted N3C Secure Data Enclave, which integrates EHR data for 18 million patients from 76 participating sites across the United States. N3C's methods for patient data acquisition, ingestion, and harmonization have been reported in detail elsewhere [23–25]. All analyses and results as part of this study are reported in adherence with the Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) reporting guidelines (S1 STROBE Checklist) [26].

# Eligibility criteria

We defined our study period as April 1, 2022 to August 28, 2023, with an index cutoff date of July 31, 2023. We excluded the period between December 21, 2021 (the date of Paxlovid's EUA) and March 31, 2022, due to low treatment rates during this period and to make the study more relevant to the current phase of the pandemic. To meet the eligibility criteria for the study as per the target trial protocol, we specified the following inclusion criteria: (1) having

| T.LL. 1   | Durste sell of a transformer that a setting of a the offerst of a selle |                 | · · · · · · · · · · · | • 11         | 1. 4L . 20 J   | C 11              | CADC C     | - X7 3 4 4    |
|-----------|---|-----------------|-----------------------|--------------|----------------|-------------------|------------|---------------|
| I adle I. | Protocol of a target trial to estimate the effect of baxio              | ovid on the rai | te of nosdi           | talization i | in the 28 days | following a posit | ive SARS-U | 0 V - 2 test. |
|           |   |                 |                       |              |                |                   |            |               |

| Protocol<br>component    | Description under target trial conditions   | Method of target trial emulation in this study  |  |  |  |
|--------------------------|---|---|--|--|--|
| Eligibility<br>criteria  | Persons aged 18 years and older, who are not currently hospitalized and<br>who have an acute COVID-19 infection, and are eligible for Paxlovid<br>treatment due to presence of one or more risk factors for severe COVID-<br>19 as per Center for US Centers for Disease and Prevention (CDC)<br>guidelines [20].                                       | Persons aged 18 years and older who are not currently hospitalized, who<br>have one or more risk factors for severe COVID-19 as per CDC<br>guidelines documented in their EHR and were not prescribed a drug<br>with a severe interaction with Paxlovid in the 30 days prior to index, with<br>a COVID-19 index (either a documented COVID-19 diagnosis or<br>positive SARS-CoV-2 lab test) during the study period [27]. |  |  |  |
| Treatment<br>strategies  | nent<br>giesPaxlovid prescribed within a 5-day grace period of the date the patient<br>presented with acute COVID-19.Paxlovid prescribed within 5 days of COVID-19 index, i<br>Paxlovid or nirmatrelvir drug exposure record in their E<br>days of the COVID-19 diagnosis or positive SARS-CoV-<br>constitutes their COVID-19 index date for the study. |   |  |  |  |
| Assignment<br>procedures | Participants will be randomly assigned to treatment or control at the date they present with acute COVID-19 and will be aware of their treatment assignment.  | Participants will be cloned across both treated and untreated arms, and<br>will be assigned IPCW corresponding to their probability of remaining<br>uncensored through the study period, to maintain the comparability of<br>the study arms throughout the study period.  |  |  |  |
| Follow-up<br>period      | Each patient will be followed for 28 days after treatment assignment.<br>Patients who obtain Paxlovid from an outside source will be censored.  | Patients will be censored at 28 days following treatment assignment<br>across both groups post-cloning. Patients in the control group that<br>deviate from their assigned treatment will be censored if they receive<br>Paxlovid during the study period.   |  |  |  |
| Outcome                  | Hospitalization within follow-up period   | Hospitalization within follow-up period   |  |  |  |
| Causal contrasts         | Per-protocol effect   | Per-protocol effect   |  |  |  |
| Analysis plan            | Estimation of effects via comparison of hospitalization rates among<br>individuals assigned to each treatment arm, adjusting for pre-treatment<br>covariates. Additional analyses to include the effects of vaccination and<br>effect heterogeneity across age strata.  | Estimation of cumulative incidence of hospitalization in each treatment<br>arm using IPC-weighted Kaplan–Meier estimators; estimate the relative<br>risk and absolute risk differences based on point estimates and variances<br>of cumulative incidence estimates. Additional analyses to include the<br>effects of vaccination, and effect heterogeneity across age strata.   |  |  |  |

COVID-19, Coronavirus Disease 2019; EHR, electronic health record; IPCW, inverse probability of censoring weights; SARS-CoV-2, Severe Acute Respiratory Syndrome Coronavirus 2.

https://doi.org/10.1371/journal.pmed.1004493.t001

a documented COVID-19 index date within the study period (with index date defined as the earliest date of either (a) COVID-19 diagnosis or (b) positive SARS-CoV-2 test result); (2) being  $\geq$ 18 years of age as of the COVID-19 index date; (3) indicated for Paxlovid treatment by having one or more risk factors for severe COVID-19 as per CDC guidelines, including age  $\geq$ 50 years old, or the presence of underlying medical conditions associated with a higher risk of severe COVID-19 [27]. We excluded all patients <18 years due to the potential for differences in both clinical characteristics and prescription practices among pediatric and adult patients [28,29].

We also specified 3 exclusion criteria: (1) patients who were hospitalized on or before the COVID-19 index date or date of treatment with Paxlovid (outcome precluding treatment); (2) patients who received Paxlovid before their COVID-19 index date; (3) patients who were prescribed a drug with a severe interaction with Paxlovid in the 30 days prior to the COVID-19 index [30]. In order to ensure that data were captured from sites with high fidelity and adequate coverage, we only included data from sites with at least 1% of eligible patients, and a minimum of 100 patients, treated with Paxlovid during the study period. This site-level exclusion criterion removed sites where Paxlovid was effectively unused, improving positivity across treatment arms.

#### Treatment and outcome

Eligible patients were categorized by their treatment exposure, defined as receiving a Paxlovid prescription within a 5-day grace period of their COVID-19 index date, with controls defined as patients not prescribed Paxlovid in that period. We selected a 5-day grace period to parallel the clinical guideline to treat within 5 days of symptom onset and to minimize heterogeneity and potential for indication bias. Additionally, specifying a grace period and adjusting for this analytically allows us to avoid poorly defined interventions that may introduce errors in causal inference [31]. For patients who were never treated with Paxlovid, we used the date of their earliest indication of COVID-19 (diagnosis or positive lab result) within the study period as the index date. For those in the treatment group, we used their earliest indication of COVID-19 within 5 days of their first Paxlovid treatment date in the study period. Within the N3C enclave, the "Paxlovid or nirmatrelvir" concept set was used to identify drug exposures that correspond to Paxlovid (10 Observational Medical Outcomes Partnership [OMOP] concepts) [32]. We followed patients for a 28-day period following their COVID-19 index date. Our primary outcome was hospitalization at any point during the 28-day follow-up period. We also assessed death and a composite outcome of death or hospitalization as additional outcomes.

### **Descriptive analysis**

First, we applied two-sided Chi-squared tests to examine the distribution of Paxlovid treatment across 2 covariates: (1) patient race and ethnicity; and (2) a ZIP code-level community well-being index (hereafter referred to as CWBI). The CWBI measure is a composite index of social determinants of health available within the N3C database, with higher CWBI values corresponding to a higher level of protective community-level social determinants of health. The index methodology was developed by Sharecare and the Boston University School of Public Health. CWBI values are derived from the patient's residential ZIP code-level data across 5 key inter-related community-level domains: healthcare access (ratios of healthcare providers to population), resource access (libraries and religious institutions, employment, and grocery stores), food access (access to grocery stores and produce), housing and transportation (home values, ratio of home value to income, and public transit use), and economic security (rates of employment, labor force participation, health insurance coverage rate, and household income above the poverty level) [33].

#### Cloning, censoring, and weighting

Next, we used the potential outcomes framework to estimate the effect of Paxlovid treatment on hospitalization. We applied the clone-censor-weight technique, which is well-suited when the treatment grace period overlaps the outcome observation period [21,34]. This creates a period of "immortal time" for patients in the treatment group because they inherently survived without having the outcome until the time of treatment. This immortal time may bias estimates in favor of a lower risk in the treatment group, even in the absence of a true effect [31].

The clone-censor-weight technique assumes that, at time zero, all eligible patients could potentially be treated or not. We defined time zero ( $t_0$ ) as the date of a patient's COVID-19 index date and  $t_A$  as the time of treatment assignment. For patients in the treatment group,  $t_A$ is the date of Paxlovid prescription within the 5-day grace period. For patients in the control group,  $t_A$  is day 5, when the grace period is over and treatment is no longer possible. We created a clone of each patient in our study cohort, with the clone assigned to the opposite treatment group of the true patient. Assigning each patient to both treatment strategies that are compatible with their observed data at  $t_0$  eliminates immortal time bias because the period between  $t_0$  and  $t_A$  counts as both treated and untreated [34,35]. It also makes the 2 treatment groups identical in pretreatment characteristics because they contain the same set of patients. This establishes exchangeability and removes confounding bias at  $t_0$ .

Next, we applied artificial censoring to ensure that patients and clones follow their assigned treatment strategy after  $t_0$ . Clones were censored at the time of their true patient counterpart's  $t_A$ . Therefore, clones of treated patients were censored from the control arm at the date of their true counterpart's treatment. Clones of control patients were censored from the treated arm at day 5 when the grace period ended. In addition to this artificial censoring, for the hospitalization outcome, we treated death as a censoring event if it occurred posttreatment assignment, pre-outcome, within the study period.

Although the clone-censor-weight technique removes immortal time bias, it also introduces selection bias through informative artificial censoring [36]. To adjust for this selection bias, we applied inverse probability of censoring weights (IPCW) [37,38]. IPCW up-weights patients who remain at risk during the study period, such that the weighted cohort is equivalent to a cohort in which censoring is random. To calculate IPCW, we used logistic regression models to estimate the probability of remaining at risk at each day of the grace period, conditional on remaining at risk the previous day and on a set of covariates selected a priori based on clinical relevance to treatment and outcome. To convert daily probabilities to probabilities of remaining at risk through each day, we took the cumulative product of the daily probabilities. We stabilized the inverse of these probabilities and trimmed them to the 1st and 99th percentile. The value from day 5 (when all clones had been artificially censored) was carried forward through the rest of the study period.

IPCW covariates included sex, age (binned), race and ethnicity, prior history of individual comorbid conditions captured in the Charlson comorbidity index (CCI), value of the composite CCI (binned), prior history of conditions associated with risk of severe COVID-19 (as defined by the CDC Paxlovid eligibility criteria), CWBI (binned), number of visits in the year prior to index (binned), number of hospitalizations in the year prior to index (binned), month of COVID-19 onset, and site of care provision [27]. We included race, ethnicity, and CWBI as potential confounders due to the prevalence of literature suggesting disparity in treatment assignment and outcomes by these variables [17–20]. Sex, age, and comorbidities are known to affect both care-seeking and the outcome of COVID-19. The index month was included because Paxlovid treatment rates, viral variants, and infection rates changed during the study period. CCI was coded as missing when no condition exposures were present in N3C prior to index. CWBI was coded as missing when patient ZIP code was not reported.

#### Effect estimation

We conducted an IPC-weighted analysis to estimate the effect of receiving Paxlovid (treatment), compared to not receiving Paxlovid (control), on the outcomes of hospitalization, death, and hospitalization, or death, among patients with acute COVID-19. Specifically, we estimated the cumulative incidence of each outcome in the 28 days following a COVID-19 infection using an IPC-weighted Kaplan–Meier estimator [38]. The weighted Kaplan–Meier estimator estimates the survival function as follows:

$$\hat{S}_{\rm IPCW}(t) = \begin{cases} 1 & \text{if } t < t_1 \\ \prod_{t_i \le st} \left[ 1 - \frac{\sum_{j: \delta_j(t_i) = 1} \hat{W}_{j(t_i)}}{\sum_{k=1}^n R_k(t_i) \hat{W}_{k(t_i)}} \right] & \text{if } t \ge t_1 \end{cases}$$

Where:

- $\hat{S}_{IPCW}(t)$  is the weighted estimated survival function at time *t*.
- *t<sub>i</sub>* are the event times.
- $\delta_j(t_i)$  is an indicator function (1 if individual *j* experiences the event at time  $t_i$ , 0 otherwise).
- $\hat{W}_{i(t_i)}$  is the estimated weight for individual *j* at time  $t_i$ .
- $R_k(t_i)$  is an indicator function (1 if individual k is at risk at time  $t_i$ , 0 otherwise).
- *n* is the total number of individuals.

We took the complement of the survival function as the cumulative incidence function. We estimated confidence intervals using a nonparametric bootstrap with 200 iterations. We present both absolute and relative risk estimates, and estimate number needed to treat (the number of patients who need to be treated to achieve 1 additional favorable outcome) as the reciprocal of the estimated absolute risk difference between treatment arms within the follow-up period [39–41].

#### Vaccination-adjusted sub-analysis and stratified analysis

Our primary analysis did not consider COVID-19 vaccination status because vaccination information is subject to missingness in most EHRs. However, we conducted a vaccination-sub-analysis in a subset of sites with high-quality vaccination data. We used the same method applied to the primary analysis described above with an additional indicator of whether or not the patient was fully vaccinated at least 2 weeks prior to index included as a covariate. We considered this important for 2 reasons. First, we hypothesized that vaccination status may be a confounder of Paxlovid treatment and hospitalization, largely through the latent variable of infection severity or propensity to seek care. If so, our primary analysis would violate the assumption of no unmeasured confounding. Second, we hypothesized that Paxlovid treatment may be less effective among vaccinated patients. There is mixed evidence for this dynamic. Pfizer's Evaluation of Protease Inhibition for COVID-19 in Standard-Risk Patients (EPIC-SR) trial found no significant treatment effect among vaccinated patients with one risk factor for severe COVID-19 [42,43]. A large health system found that Paxlovid treatment reduced the likelihood of hospitalization and death among vaccinated patients, but that the treatment effect was smaller than among unvaccinated patients [10].

For this sub-analysis, we used a modified cohort of patients from sites with reliable information on patient vaccination status. Vaccination status in N3C is subject to misclassification, particularly among patients who were vaccinated outside of data partner systems. We determined the subset of sites with reliable patient-level vaccination records by the ratio of 2 statistics: (1) the proportion of individuals who are documented as vaccinated in their EHR; and (2) the proportion of individuals who are truly vaccinated. We calculated the first statistic from the EHR data for each site. We estimated the second statistic using CDC-reported vaccination rates for the counties served by each partner facility [44]. CDC vaccination rates by county are included as a data asset in N3C. Patient counties were inferred using a county-ZIP crosswalk. Each patient's likelihood of vaccination was drawn from their county's vaccination rate and the overall expected vaccination rate for a partner facility was computed as the mean of their patients' vaccination likelihood. We defined a facility's recorded vaccine ratio as the ratio of these 2 statistics and limited the vaccination sub-analysis to individuals from facilities with a ratio of at least 0.66 [45].

Accordingly, we categorized patients by their vaccination status prior to their COVID-19 index date, defined as having completed a full course of vaccination at least 14 days prior to index. Partially vaccinated patients and patients who were fully vaccinated fewer than 14 days prior to index were excluded from the analysis. We estimated the absolute and relative treatment effect of Paxlovid on hospitalization in this cohort using the same statistical analysis steps described for the primary analysis.

We also conducted a stratified analysis to examine effect heterogeneity by vaccination status. For fully vaccinated and unvaccinated patients in the vaccination-aware cohort, we separately estimated the absolute and relative treatment effect of Paxlovid on hospitalization using the same statistical analysis steps described for the primary analysis.

# Age-stratified analysis

We conducted a second stratified analysis to examine effect heterogeneity across 3 strata of patient age at COVID-19 index: 18 to 49 years, 50 to 64 years, and 65+ years. We selected these strata due to their clinical and policy implications. The first cutoff, age 50 years, was grounded in the eligibility criteria for treatment with Paxlovid. Patients below 50 years old are not indicated for treatment unless they have additional risk factors for severe COVID-19. We chose the second cutoff of age of 65 years due to its policy relevance as an eligibility cutoff for health coverage through Medicare, and consequent increase in utilization of services [46].

# **Ethics** approval

The N3C data transfer to the National Center for Advancing Translational Sciences is performed under a Johns Hopkins University reliance protocol (IRB00249128). The N3C Publications Committee approved this manuscript (manuscript ID 1020.56). Results downloads were approved per N3C Attribution and Publication Principles [47].

# Results

The hypothesized target trial protocol is articulated in <u>Table 1</u>, and we emulated each component to define our cohort within the N3C database (see Fig 1).

# Patient characteristics

In the study cohort (within our defined study period), a total of 703,647 patients had a valid COVID-19 index date during the study period, of which 206,393 (20.6%) were treated with Paxlovid, and a total of 13,895 (2.0%) patients across the entire study cohort were hospitalized. After applying the eligibility criteria to the patient population and selecting high-fidelity Paxlovid prescription sites, a total of 34 of 76 study sites were retained. The characteristics of all patients during the study period are presented in Table 2, stratified by treatment group.



Fig 1. Study cohort and flow of emulated trial. COVID-19, Coronavirus Disease 2019; SARS-CoV-2, Severe Acute Respiratory Syndrome Coronavirus 2.

https://doi.org/10.1371/journal.pmed.1004493.g001

Among the study cohort, there were large, statistically significant differences in assignment to Paxlovid treatment. In our study cohort, 15.7% of black, non-Hispanic patients and 15.1% of Hispanic patients were treated with Paxlovid, compared to 22.4% of white, non-Hispanic patients, and 23.1% of Asian, non-Hispanic patients ( $X^2_{df=5}$  test of independence *p*-value <0.001). When stratified by patients' residential areas, patients who lived in areas with higher CWBI values (and lower corresponding social vulnerability) were also more likely to be treated with Paxlovid ( $X^2_{df=4}$  test of independence *p*-value <0.001; see Fig 2). The relationship between race and ethnicity, and CWBI values across the study cohort of included patients is provided in S1 Fig.

# Effect of Paxlovid on hospitalization

In our study cohort, hospitalization rates differed markedly between Paxlovid-treated and untreated patients. Among those treated with Paxlovid, 1,901 (0.9%) were hospitalized during the follow-up period, compared to 11,994 (2.4%) in the untreated group. Our primary analysis,

| Table 2. Baseline population characteristics.      |                       |                       |  |  |
|--|-----------------------|-----------------------|--|--|
| Characteristic                                     | Treatment group       |                       |  |  |
|  | No Paxlovid           | Paxlovid              |  |  |
|  | ( <i>N</i> = 497,085) | ( <i>N</i> = 206,562) |  |  |
| Hospitalization (within 28-day observation period) |                       |                       |  |  |
| No   | 485,091 (97.6%)       | 204,661 (99.1%)       |  |  |
| Yes  | 11,994 (2.4%)         | 1,901 (0.9%)          |  |  |
| Sex  |                       |                       |  |  |
| Female   | 312,104 (62.8%)       | 127,854 (61.9%)       |  |  |
| Male   | 184,903 (37.2%)       | 78,670 (38.1%)        |  |  |
| Missing  | 78 (0.0%)             | 38 (0.0%)             |  |  |
| Age (in years)                                     |                       |                       |  |  |
| 18-24  | 30,332 (6.1%)         | 4,722 (2.3%)          |  |  |
| 25-34  | 65,971 (13.3%)        | 14,825 (7.2%)         |  |  |
| 35-49  | 102,845 (20.7%)       | 37,479 (18.1%)        |  |  |
| 50-64  | 141,569 (28.5%)       | 66,275 (32.1%)        |  |  |
| 65+  | 156,368 (31.5%)       | 83,261 (40.3%)        |  |  |
| Race and ethnicity                                 |                       |                       |  |  |
| Asian non-Hispanic                                 | 20,391 (4.1%)         | 9,374 (4.5%)          |  |  |
| Black or African American non-Hispanic             | 68,292 (13.7%)        | 20,301 (9.8%)         |  |  |
| Hispanic or Latino any Race                        | 49,826 (10.0%)        | 16,000 (7.7%)         |  |  |
| White non-Hispanic                                 | 316,839 (63.7%)       | 147,751 (71.5%)       |  |  |
| Other non-Hispanic                                 | 8,123 (1.6%)          | 2,359 (1.1%)          |  |  |
| Unknown  | 33,614 (6.8%)         | 10,777 (5.2%)         |  |  |
| CCI2   |                       |                       |  |  |
| 0  | 249,618 (50.2%)       | 101,444 (49.1%)       |  |  |
| 1-2  | 132,010 (26.6%)       | 66,593 (32.2%)        |  |  |
| 3-4  | 42,029 (8.5%)         | 18,421 (8.9%)         |  |  |
| 5-10   | 25,557 (5.1%)         | 8,821 (4.3%)          |  |  |
| 11+  | 2,798 (0.6%)          | 820 (0.4%)            |  |  |
| Missing  | 45,073 (9.1%)         | 10,463 (5.1%)         |  |  |
| Number of clinical visits in prior year            |                       |                       |  |  |
| 0  | 70,533 (14.2%)        | 21,843 (10.6%)        |  |  |
| 1-3  | 84,908 (17.1%)        | 23,641 (11.4%)        |  |  |
| 4-9  | 102,173 (20.6%)       | 41,641 (20.2%)        |  |  |
| 10-20  | 107,292 (21.6%)       | 54,840 (26.5%)        |  |  |
| More than 20                                       | 132,179 (26.6%)       | 64,597 (31.3%)        |  |  |
| Number of hospitalizations in prior year           |                       |                       |  |  |
| 0  | 465,424 (93.6%)       | 197,380 (95.6%)       |  |  |
| 1 visit  | 23,513 (4.7%)         | 7,328 (3.5%)          |  |  |
| More than 1 visit                                  | 8,148 (1.6%)          | 1,854 (0.9%)          |  |  |
| CWBI3  |                       |                       |  |  |
| 0-45   | 4,082 (0.8%)          | 1,241 (0.6%)          |  |  |
| 46-55  | 185,995 (37.4%)       | 65,984 (31.9%)        |  |  |
| 56-65  | 223,789 (45.0%)       | 97,878 (47.4%)        |  |  |
| 65+  | 30,591 (6.2%)         | 19,777 (9.6%)         |  |  |
| Missing  | 52,628 (10.6%)        | 21,682 (10.5%)        |  |  |
| SARS-CoV-2 test type                               |                       |                       |  |  |
| Antibody   | 498 (0.1%)            | 38 (0.1%)             |  |  |
|  |                       | (Continued)           |  |  |

### PLOS MEDICINE

Table 2. (Continued)

| Tuble 2. (Commuted)  |                       |                       |  |  |
|--|-----------------------|-----------------------|--|--|
| Characteristic   | Treatment group       |                       |  |  |
|  | No Paxlovid           | Paxlovid              |  |  |
|  | ( <i>N</i> = 497,085) | ( <i>N</i> = 206,562) |  |  |
| Antigen  | 71,846 (16.0%)        | 18,318 (17.9%)        |  |  |
| Molecular (polymerase chain reaction and nucleic acid amplification tests) | 424,199 (79.2%)       | 54,623 (77.9%)        |  |  |
| Other  | 27,910 (4.7%)         | 1,877 (2.3%)          |  |  |
| Deaths within 28-day observation period                                    |                       |                       |  |  |
| No   | 495,773 (99.7%)       | 206,435 (99.9%)       |  |  |
| Yes  | 1,312 (0.3%)          | 127 (0.1%)            |  |  |
| Intermittent mandatory ventilation during hospitalization                  |                       |                       |  |  |
| No   | 496,810 (99.9%)       | 206,511 (100.0%)      |  |  |
| Yes  | 275 (0.1%)            | 51 (0.0%)             |  |  |
| Month of COVID-19 diagnosis  |                       |                       |  |  |
| Apr 2022   | 25,584 (5.1%)         | 5,088 (2.5%)          |  |  |
| May 2022   | 60,027 (12.1%)        | 17,910 (8.7%)         |  |  |
| Jun 2022   | 62,628 (12.6%)        | 20,808 (10.1%)        |  |  |
| Jul 2022   | 70,240 (14.1%)        | 26,357 (12.8%)        |  |  |
| Aug 2022   | 58,862 (11.8%)        | 21,618 (10.5%)        |  |  |
| Sep 2022   | 38,479 (7.7%)         | 14,827 (7.2%)         |  |  |
| Oct 2022   | 29,224 (5.9%)         | 11,297 (5.5%)         |  |  |
| Nov 2022   | 27,750 (5.6%)         | 12,895 (6.2%)         |  |  |
| Dec 2022   | 38,098 (7.7%)         | 22,070 (10.7%)        |  |  |
| Jan 2023   | 28,537 (5.7%)         | 15,140 (7.3%)         |  |  |
| Feb 2023   | 18,970 (3.8%)         | 12,044 (5.8%)         |  |  |
| Mar 2023   | 14,148 (2.8%)         | 9,392 (4.5%)          |  |  |
| Apr 2023   | 8,733 (1.8%)          | 5,879 (2.8%)          |  |  |
| May 2023   | 6,651 (1.3%)          | 4,632 (2.2%)          |  |  |
| Jun 2023   | 4,758 (1.0%)          | 3,251 (1.6%)          |  |  |
| Jul 2023   | 4,396 (0.9%)          | 3,354 (1.6%)          |  |  |

<sup>1</sup>Any hospital admission in the 28-days following a positive SARS-CoV-2 test result.

<sup>2</sup>CCI is a weighted index to predict risk of death within 1 year of hospitalization for patients with specific comorbid conditions, and this implementation uses the weights from Quan and colleagues [48,49].

<sup>3</sup>CWBI is a measure of 5 interrelated community-level domains: Healthcare access (ratios of healthcare providers to population), resource access (libraries and religious institutions, employment, and grocery stores), food access (access to grocery stores and produce), housing and transportation (home values, ratio of home value to income, and public transit use), and economic security (rates of employment, labor force participation, health insurance coverage rate, and household income above the poverty level) [33].

CCI, Charlson comorbidity index; CWBI, community well-being index; SARS-CoV-2, Severe Acute Respiratory Syndrome Coronavirus 2.

https://doi.org/10.1371/journal.pmed.1004493.t002

utilizing a cloned, censored, and IPC-weighted cohort, revealed that Paxlovid treatment was associated with a significantly lower relative risk (RR, 0.614; 95% CI [0.593, 0.636];  $p \le 0.001$ ). Additionally, the estimated absolute risk difference (ARD) indicated a substantive absolute risk reduction in the treated cohort (ARD, 0.009; 95% CI [0.009, 0.010];  $p \le 0.001$ ), corresponding to a number needed to treat (NNT) of 111 to avert 1 hospitalization during the follow-up period (see Fig 3 and Table 3).

# Effect of Paxlovid on death and composite outcome

We also evaluated the effect of Paxlovid on mortality and a composite outcome of hospitalization and death (see Fig 3 and Table 3). Among Paxlovid-treated patients, 127 (0.1%) died



Notes: Abbreviations: NH = Non-Hispanic; Lower CWBI scores indicate higher community-level social vulnerability Fig 2. Paxlovid treatment rates by patient race and ethnicity and CWBI. CWBI, community well-being index.

https://doi.org/10.1371/journal.pmed.1004493.g002

during follow-up, compared to 1,312 (0.3%) in the untreated group. Paxlovid treatment was associated with a significantly lower relative risk of death (RR, 0.389; 95% CI [0.333, 0.446];  $p \le 0.001$ ) and a small but meaningful absolute risk reduction (ARD, 0.002; 95% CI [0.001, 0.002];  $p \le 0.001$ ), corresponding to an NNT of 500 to avert 1 death during the follow-up period. For the composite outcome of hospitalization and death, Paxlovid treatment similarly showed both a lower relative risk (RR, 0.597; 95% CI [0.577, 0.617];  $p \le 0.001$ ) and a



Note: All analyses report Relative Risk (RR) and 95% Confidence Intervals (CI). Vaccination-adjusted analysis replicates the analysis examining the effect of Paxlovid on hospitalization as our main outcome, controlling for the effect of being fully vaccinated, relative to being unvaccinated. 'Fully vaccinated' included only patients that completed a full course (2 or more doses) of vaccination at least 14 days prior to index, and 'unvaccinated patients' included only patients who received no vaccinations prior to index. Partially vaccinated patients and patients who were fully vaccinated fewer than 14 days prior to index were excluded from the analysis. For Vaccination-stratified analysis, our study cohort used a modified cohort of patients from clinical sites with reliable information on patient vaccination.

Fig 3. Forest plot of treatment effects of Paxlovid on patients with COVID-19 across all analyses (reported as relative risks). CI, confidence interval; COVID-19, Coronavirus Disease 2019; RR, relative risk.

https://doi.org/10.1371/journal.pmed.1004493.g003

significant absolute risk reduction (ARD, 0.010; 95% CI [0.010, 0.011];  $p \le 0.001$ ), corresponding to an NNT of 100 to avert 1 hospitalization or death during the follow-up period. These results demonstrate that Paxlovid not only reduces the risk of hospitalization but also significantly lowers mortality risk, with the composite outcome reinforcing its overall clinical benefit.

### Effect of Paxlovid by COVID-19 vaccination status

For our subgroup analysis estimating the effect of Paxlovid on hospitalization, we examined both vaccination status and age groups (see Fig 3 and Table 3). The vaccination analysis included 272,080 vaccinated individuals (29.1% received Paxlovid, 1.3% hospitalized) and 106,832 unvaccinated individuals (14.1% received Paxlovid, 2.3% hospitalized) from 9 sites with trusted vaccine data. After cloning, censoring, and adjusting for vaccination status, the overall relative risk (RR, 0.599; 95% CI [0.559, 0.639];  $p \le 0.001$ ) and absolute risk reduction (ARD, 0.007; 95% CI [0.006, 0.008];  $p \le 0.001$ ) were similar to the primary analysis, with an estimated NNT of 143 to avert 1 hospitalization during the follow-up period. Fully vaccinated and unvaccinated patients showed similar absolute risk reductions with Paxlovid (ARD, 0.007 versus 0.006), but vaccinated patients had a larger relative risk reduction (RR, 0.579 versus 0.739) due to their lower baseline hospitalization risk.

### Effect of Paxlovid by age group

In our age-stratified analysis, we observed varying effects of Paxlovid across different age groups (see Fig 3 and Table 3). Notably, while Paxlovid provides a benefit across all age groups,

| Analysis                                    | Cumulative incidence (95% CI) |                         | RR (95% CI)               | Absolute risk difference (95% CI)    | NNT2                |  |
|---|-------------------------------|-------------------------|---------------------------|--------------------------------------|---------------------|--|
|   | Paxlovid                      | No Paxlovid             |                           |                                      |                     |  |
| Main results                                |                               |                         |                           |                                      |                     |  |
| Hospitalization as the main outcome         | 0.015 (0.014,<br>0.015)       | 0.024 (0.023,<br>0.024) | 0.614 (0.593, 0.636)      | 0.009 (0.009, 0.010)*                | 111 (100, 111)      |  |
| Death as a main outcome                     | 0.001 (0.001,<br>0.001)       | 0.003 (0.003,<br>0.003) | 0.389 (0.333, 0.446)      | 0.002 (0.001, 0.002)*                | 500 (500,<br>1,000) |  |
| Death and hospitalization composite outcome | 0.015 (0.015,<br>0.016)       | 0.026 (0.025,<br>0.026) | 0.597 (0.577, 0.617)      | 0.010 (0.010, 0.011)*                | 100 (91, 100)       |  |
| Vaccination, stratified analysis1           |                               |                         |                           |                                      |                     |  |
| Vaccination-adjusted analysis               | 0.011 (0.010,<br>0.011)       | 0.018 (0.018,<br>0.019) | 0.599 (0.559, 0.639)<br>* | 0.007 (0.006, 0.008)*                | 143 (125, 167)      |  |
| Fully vaccinated patients only              | 0.009 (0.009,<br>0.010)       | 0.016 (0.016,<br>0.017) | 0.579 (0.533, 0.626)      | 0.007 (0.006, 0.008)*                | 143 (125, 167)      |  |
| Unvaccinated patients only                  | 0.017 (0.016,<br>0.019)       | 0.023 (0.022,<br>0.025) | 0.739 (0.664, 0.813)      | 0.006 (0. <mark>0</mark> 04, 0.008)* | 167 (125, 250)      |  |
| Age, stratified analysis                    |                               |                         |                           |                                      |                     |  |
| Ages 18 to 49 years                         | 0.015 (0.014,<br>0.016)       | 0.022 (0.021,<br>0.023) | 0.670 (0.632, 0.708)      | 0.007 (0.006, 0.008)*                | 143 (125, 167)      |  |
| Ages 50 to 65 years                         | 0.009 (0.009,<br>0.010)       | 0.016 (0.016,<br>0.017) | 0.564 (0.520, 0.608)      | 0.007 (0.006, 0.008)*                | 143 (125, 167)      |  |
| Ages 65+ years                              | 0.019 (0.018,<br>0.020)       | 0.033 (0.032,<br>0.033) | 0.590 (0.560, 0.621)      | 0.013 (0.012, 0.015)*                | 77 (67, 84)         |  |

#### Table 3. Estimated cumulative incidence, RRs, absolute risk reduction, and numbers needed to treat associated with treatment with Paxlovid across all analyses.

#### \*p < 0.001.

<sup>1</sup>Vaccination-adjusted analysis replicates the analysis examining the effect of Paxlovid on hospitalization as our main outcome, controlling for the effect of being fully vaccinated, relative to being unvaccinated. "Fully vaccinated" included only patients that completed a full course (2 or more doses) of vaccination at least 14 days prior to index, and "unvaccinated patients" included only patients who received no vaccinations prior to index. Partially vaccinated patients and patients who were fully vaccinated from the analysis. For Vaccination-stratified analysis, our study cohort used a modified cohort of patients from clinical sites with reliable information on patient vaccination.

 $^{2}$ NNT calculated as (1 ÷ absolute risk difference), corresponding to the expected number of patients that must be treated to prevent the outcome in one patient within the specified follow-up period, rounded up to the nearest integer.

CI, confidence interval; COVID-19, Coronavirus Disease 2019; RR, relative risk.

https://doi.org/10.1371/journal.pmed.1004493.t003

the absolute risk reduction is most pronounced in older adults, likely due to their higher baseline risk of hospitalization. For individuals aged 18 to 49 years, Paxlovid treatment was associated with a relative risk of 0.670 (95% CI [0.633, 0.708];  $p \le 0.001$ ) and an absolute risk reduction of 0.007 (95% CI [0.006, 0.008];  $p \le 0.001$ ) for hospitalization, corresponding to an NNT of 143 to avert 1 hospitalization during the follow-up period. In the 50 to 65 years age group, the relative risk was 0.564 (95% CI [0.520, 0.608];  $p \le 0.001$ ) with an absolute risk reduction of 0.007 (95% CI [0.006, 0.008];  $p \le 0.001$ ), also corresponding to an NNT of 143 to avert 1 hospitalization during the follow-up period. The greatest benefit was observed in the 65 + years age group, where Paxlovid treatment resulted in a relative risk of 0.590 (95% CI [0.560, 0.621];  $p \le 0.001$ ) and an absolute risk reduction of 0.013 (95% CI [0.012, 0.015];  $p \le 0.001$ ), corresponding to an NNT of 77 to avert 1 hospitalization during the follow-up period.

# Discussion

In this target trial emulation using the N3C database, Paxlovid treatment within 5 days of a COVID-19 diagnosis or positive SARS-CoV-2 test reduced the risk of hospitalization by 39%, death by 61%, and hospitalization or death by 40%. The effect size for hospitalization is similar

to the association found in a Cleveland Clinic study (37%), which also used a contemporary study period, and the effect size found in VA and Mass General Brigham studies (40%), which also used target trial emulation methods [9–11]. We confirm these findings using a larger, more representative cohort. Although other studies have found varying effect sizes, we believe this constitutes a growing consensus on the magnitude of Paxlovid's effect in the real world. While our findings are consistent with other real-world studies, they differ from the results of early clinical trials like EPIC-HR. This discrepancy is likely due to differences in study design rather than changes in virus variants. Factors such as inclusion criteria, timing of treatment initiation, and estimation of intention-to-treat versus per-protocol effects may account for the lower effectiveness observed in real-world settings compared to clinical trials.

The Cleveland Clinic study was widely interpreted in the media with headlines like "Paxlovid Weaker Against Current COVID-19 Variants" and "Paxlovid Now Less Effective Than In Early Trials" [8,9]. We consider that to be a misinterpretation. Earlier real-world studies found similar or lower effect sizes in study periods dominated by earlier variants [5,10,11]. Realworld studies of Paxlovid's effectiveness have also differed in population, sampling, outcomes, and methodology, and these are more likely than the dominant variant during the study period to account for their differences. It is true that real-world studies have consistently found a lower risk reduction than in the EPIC-HR trial. This may be attributable to various differences in study design (as noted by Najjar-Debbiny and colleagues): (1) differences in dominant strains at the time of the study; (2) inclusion of only symptomatic patients versus inclusion of all COVID-19 positive patients (as well as other possible cohort or methodological differences); (3) treatment assignment within 5 days of symptom onset versus treatment assignment within 5 days of a COVID-19 diagnosis or a positive SARS-CoV-2 test; and (4) estimation of the intention-to-treat effect versus the per-protocol effect [6]. We consider the latter to be the most likely hypothesis, but further research on the source of this difference is needed. Regardless, Paxlovid is not less effective now than in early trials, it is less effective in the real world than in early trials. This common misinterpretation illustrates how media reports often draw the wrong conclusions from scientific research, and why the scientific community must strive to interpret findings for lay audiences.

In a subcohort with reliable vaccination data, we found that Paxlovid treatment reduced the absolute risk of hospitalization by 0.7 percentage points, regardless of vaccination status. This reduction in absolute risk translates to an NNT of 143, meaning that for every 143 individuals treated with Paxlovid, one hospitalization would be prevented. While the relative risk reduction (RRR) for vaccinated patients was larger (42% versus 26% for unvaccinated patients), the consistent ARD across both groups suggests that the benefit of Paxlovid in preventing hospitalization is similar for both vaccinated and unvaccinated populations. Although the NNT is high, particularly in vaccinated patients with lower baseline risk, it is important to contextualize this within the broader public health system. In high-risk populations, even a modest reduction in absolute risk can have significant impacts, particularly in reducing the strain on the healthcare system. Additionally, the consistency of the absolute risk reduction regardless of vaccination and Paxlovid, rather than a multiplicative effect. This supports the use of Paxlovid as a valuable treatment option for preventing severe COVID-19 outcomes, even among vaccinated patients.

This is particularly noteworthy given the preliminary evidence from the EPIC-SR trial and other real-world studies that suggested uncertainty about Paxlovid's impact on vaccinated populations [10,42,43]. Our results indicate that Paxlovid treatment remains valuable even for vaccinated patients, providing a consistent absolute risk reduction regardless of vaccination status. Further research is needed on the effectiveness and cost-effectiveness of Paxlovid treatment in vaccinated populations. It is also worth noting that the overall effect of Paxlovid

treatment in this subcohort while controlling for vaccination status was similar to the primary analysis (40% relative risk reduction of hospitalization). This suggests that vaccination status did not cause substantial unmeasured confounding in the primary analysis.

In an age-stratified analysis, we found a larger absolute risk reduction for patients aged 65 + years (1.3 percentage points) compared to patients ages 18 to 49 years and 50 to 64 years (0.7 percentage points for both groups). This ARD translates into an NNT of 77 for patients aged 65+, implying that fewer individuals in this age group need to be treated to avert one hospitalization. Overall hospitalization risk was highest among patients aged 65+ years, followed by patients ages 18 to 49 years. Higher overall risk among patients ages 18 to 49 than among patients ages 50 to 64 years is explained by the cohort inclusion criteria. All patients over 50 years are indicated for Paxlovid treatment, and therefore eligible for inclusion, while patients under 50 years require a specific comorbidity associated with increased risk of severe COVID-19 to be indicated for treatment. This likely results in a higher-risk subcohort in this age group. Due to these differences in overall risk, the 18 to 49 years age group had the lowest RRR (33%), followed by the 65+ years group (41%) and the 50 to 64 years group (44%). Our findings suggest that older patients may have the most to gain from Paxlovid treatment, but further research is needed to understand the potential population health impact of differing treatment protocols by age group.

We also found large differences in Paxlovid treatment rates by race, ethnicity, and CWBI. Paxlovid is thus effective at reducing the risk of severe COVID-19 outcomes but is not equitably distributed. Black and Hispanic/Latino patients were less likely to receive Paxlovid treatment than white and Asian patients. Additionally, patients from communities with higher levels of social vulnerability (lower CWBI values) were less likely to receive Paxlovid treatment. Prior research from December 2021 to July 2022 documented similar disparities [20]. Our findings show that these disparities have persisted over time. These disparities in access to treatment are of particular concern due to the growing evidence base that shows that individuals in these groups are more likely to experience higher levels of COVID-19 exposure, discrimination in access to care, and severe COVID-19 health outcomes [50-55]. The reasons for these disparities are a complex constellation of factors including (and not limited to) a history of systemic discrimination and racism in treatment, lack of physical and economic resources to facilitate equal access in vulnerable communities, lack of information and knowledge about treatment options, and language barriers owing to a lack of culturally competent care [20]. Without attention, recognition, and remediation on the part of providers, public health agencies, the health system, and communities, the disproportional burden of COVID-19 will only further exacerbate existing health inequities in the US.

This study has several strengths that underscore the value of large-scale EHR repositories for advancing comparative effectiveness research. In the absence of large-scale randomized controlled trials, target trial emulations allow researchers to explore treatment effects in real-world settings and identify the most effective treatments for a variety of health conditions. This study is one of a few studies that apply methods to emulate hypothetical target trials, accounting for the effect of confounding [56–59]. Additionally, the analyses were conducted using a large, comprehensive database of EHR data from 34 contributing sites across the US, increasing generalizability, and decreasing the potential for issues that typically arise from misclassification in administrative or claims data [60]. We used a contemporary study period, which makes our findings more relevant to current SARS-CoV-2 subvariants than many prior studies on the topic. Finally, to our knowledge, our stratified analyses contribute novel findings on Paxlovid's heterogeneous effects by age group and homogeneous effects by vaccination status, providing valuable insights for targeted treatment strategies.

This study also has several limitations. First, since the data did not include any information on adherence (Paxlovid is typically given orally twice daily for 5 days), we were unable to account for adherence. Second, the sub-analysis on vaccinations did not include individuals with incomplete courses of vaccination (1 dose), nor information on the timing of vaccination relative to COVID-19 infection, and therefore we were unable to shed light on whether the response to Paxlovid varied by additional strata of vaccination. Third, it is well-documented that EHRs are susceptible to missing data when patients do not seek care, care is provided outside of the reporting facility, or a condition is documented outside of the structured EHR (e.g., in clinical notes), and it is likely that our estimates may be biased if missingness was related to any residual unobserved confounding [61-63]. We are also unable to assess the reason for hospitalization, as N3C does not have information on the priority ranking of discharge diagnoses. We took several steps to mitigate the risk of missing data—all individuals in our cohort have established care at the partner facility before their COVID-19 index, as evidenced by documented healthcare encounters, and our vaccination sub-analysis is limited to facilities with a high recorded vaccine ratio to reduce the number of individuals misclassified as unvaccinated. Related to this inclusion criteria, there may have been site-level variation in Paxlovid treatment rates across sites, such as differences in institutional protocols, local COVID-19 prevalence, patient demographics, and healthcare system capacity—while we may not have been able to fully account for these factors within the scope of our analysis, future work may benefit from addressing factors that may introduce site-level confounding in treatment assignment. Fourth, our inclusion criteria of Paxlovid treatment within 5 days of COVID-19 index differs from the indication of treatment within 5 days of symptom onset. However, we note that within our base cohort, 85.1% of treated patients were treated within 1 day of COVID-19 index. Fifth, this study's eligibility criteria include indication for on-label Paxlovid treatment (i.e., at risk for developing severe COVID-19 due to the presence of 1 or more risk factors). Therefore, results can only be generalized to a high-risk population. The effect of Paxlovid treatment among lower-risk patients is an important area for future research, although studying it in the real world is difficult due to confounding by indication. Finally, our study is subject to the assumptions required of all causal inference studies: consistency, positivity, and exchangeability. In particular, the assumption of exchangeability rests on the assumption that there are no unmeasured confounders. Although we controlled for a wide range of potential confounders, certain possible confounders were unmeasurable in the EHR. Patient insurance status and economic well-being could affect decisions to seek Paxlovid and hospitalization. Provider behavior related to offering Paxlovid treatment also varies in this cohort, as shown by the variation in treatment rates by site. Treatment availability may also contribute to that variation. Ultimately, although we are confident that we controlled for confounding to the greatest extent possible, residual unmeasured confounding is possible.

We found that Paxlovid treatment within 5 days of a COVID-19 diagnosis or positive test reduced the risk of hospitalization by 39%. Broadly speaking, our findings are consistent with the evidence base: Paxlovid is effective at preventing severe COVID-19 outcomes, but it is less effective in real-world settings than in early clinical trials. We used a study period (April 2022 to August 2023) that included COVID-19 cases caused mostly by SARS-CoV-2 Omicron subvariants, which suggests that Paxlovid remains effective against more recent variants. We found that Paxlovid treatment reduced the absolute risk of hospitalization by the same amount (0.7 percentage points), regardless of vaccination status, which suggests that Paxlovid treatment is valuable even for vaccinated patients. We also found that Paxlovid had the greatest absolute risk reduction among patients aged 65+ years (1.3 percentage points), compared to patients ages 18 to 49 years and 50 to 64 years (0.7 percentage points for both groups), which suggests that older patients may have the most to gain from Paxlovid treatment. Finally, we

found disparities in the rates of Paxlovid treatment. Black patients, Hispanic or Latino patients, and patients living in more vulnerable communities were treated with Paxlovid at a significantly lower rate than others. Taking action to remediate these disparities will equalize the opportunity for all high-risk patients to prevent severe COVID-19 outcomes.

# Ethics approval and consent to participate

The N3C data transfer to NCATS is performed under a Johns Hopkins University Reliance Protocol #IRB00249128 or individual site agreements with NIH. The N3C Data Enclave is managed under the authority of the NIH; information can be found at https://ncats.nih.gov/n3c/resources. The work was performed under DUR RP-5677B5.

# **Supporting information**

S1 STROBE Checklist. STROBE Statement—Checklist of items that should be included in reports of cohort studies. (DOCX)

S1 Fig. Proportion of individuals in base population stratified by race and ethnicity, and ZIP code-level community wellbeing index (CWBI). (DOCX)

# Acknowledgments

Authorship was determined using ICMJE recommendations.

The analyses described in this publication were conducted with data or tools accessed through the NCATS N3C Data Enclave covid.cd2h.org/enclave and supported by CD2H - The National COVID Cohort Collaborative (N3C) IDeA CTR Collaboration 3U24TR002306-04S2 NCATS U24 TR002306. This research was possible because of the patients whose information is included within the data from participating organizations (covid.cd2h.org/dtas) and the organizations and scientists (covid.cd2h.org/duas) who have contributed to the on-going development of this community resource [https://doi.org/10.1093/jamia/ocaa196].

We gratefully acknowledge the following core contributors to N3C: Adam B. Wilcox, Adam M. Lee, Alexis Graves, Alfred (Jerrod) Anzalone, Amin Manna, Amit Saha, Amy Olex, Andrea Zhou, Andrew E. Williams, Andrew Southerland, Andrew T. Girvin, Anita Walden, Anjali A. Sharathkumar, Benjamin Amor, Benjamin Bates, Brian Hendricks, Brijesh Patel, Caleb Alexander, Carolyn Bramante, Cavin Ward-Caviness, Charisse Madlock-Brown, Christine Suver, Christopher Chute, Christopher Dillon, Chunlei Wu, Clare Schmitt, Cliff Takemoto, Dan Housman, Davera Gabriel, David A. Eichmann, Diego Mazzotti, Don Brown, Eilis Boudreau, Elaine Hill, Elizabeth Zampino, Emily Carlson Marti, Emily R. Pfaff, Evan French, Farrukh M Koraishy, Federico Mariona, Fred Prior, George Sokos, Greg Martin, Harold Lehmann, H<mark>eidi</mark> Spratt, <mark>He</mark>malkumar Mehta, Hongfang Liu, Hythem Sidky, J.W. Awori Hayanga, Jami Pincavitch, Jaylyn Clark, Jeremy Richard Harper, Jessica Islam, Jin Ge, Joel Gagnier, Joel H. Saltz, Joel Saltz, Johanna Loomba, John Buse, Jomol Mathew, Joni L. Rutter, Julie A. McMurry, Justin Guinney, Justin Starren, Karen Crowley, Katie Rebecca Bradwell, Kellie M. Walters, Ken Wilkins, Kenneth R. Gersing, Kenrick Dwain Cato, Kimberly Murray, Kristin Kostka, Lavance Northington, Lee Allan Pyles, Leonie Misquitta, Lesley Cottrell, Lili Portilla, Mariam Deacy, Mark M. Bissell, Marshall Clark, Mary Emmett, Mary Morrison Saltz, Matvey B. Palchuk, Melissa A. Haendel, Meredith Adams, Meredith Temple-O'Connor, Michael G. Kurilla, Michele Morris, Nabeel Qureshi, Nasia Safdar, Nicole Garbarini, Noha Sharafeldin, Ofer Sadan, Patricia A. Francis, Penny Wung Burgoon, Peter Robinson, Philip R.O. Payne,

Rafael Fuentes, Randeep Jawa, Rebecca Erwin-Cohen, Rena Patel, Richard A. Moffitt, Richard L. Zhu, Rishi Kamaleswaran, Robert Hurley, Robert T. Miller, Saiju Pyarajan, Sam G. Michael, Samuel Bozzette, Sandeep Mallipattu, Satyanarayana Vedula, Scott Chapman, Shawn T. O'Neil, Soko Setoguchi, Stephanie S. Hong, Steve Johnson, Tellen D. Bennett, Tiffany Callahan, Umit Topaloglu, Usman Sheikh, Valery Gordon, Vignesh Subbian, Warren A. Kibbe, Wenndy Hernandez, Will Beasley, Will Cooper, William Hillegass, and Xiaohan Tanner Zhang.

Details of contributions available at covid.cd2h.org/core-contributors

We would also like to acknowledge our data partners: The following institutions whose data is released or pending:

Available: Advocate Health Care Network—UL1TR002389: The Institute for Translational Medicine (ITM) • Boston University Medical Campus—UL1TR001430: Boston University Clinical and Translational Science Institute • Brown University-U54GM115677: Advance Clinical Translational Research (Advance-CTR) • Carilion Clinic—UL1TR003015: iTHRIV Integrated Translational health Research Institute of Virginia • Charleston Area Medical Center—U54GM104942: West Virginia Clinical and Translational Science Institute (WVCTSI) • Children's Hospital Colorado—UL1TR002535: Colorado Clinical and Translational Sciences Institute • Columbia University Irving Medical Center-UL1TR001873: Irving Institute for Clinical and Translational Research • Duke University—UL1TR002553: Duke Clinical and Translational Science Institute • George Washington Children's Research Institute-UL1TR001876: Clinical and Translational Science Institute at Children's National (CTSA-CN) George Washington University—UL1TR001876: Clinical and Translational Science Institute at Children's National (CTSA-CN) • Indiana University School of Medicine—UL1TR002529: Indiana Clinical and Translational Science Institute • Johns Hopkins University-UL1TR003098: Johns Hopkins Institute for Clinical and Translational Research • Loyola Medicine—Loyola University Medical Center • Loyola University Medical Center— UL1TR002389: The Institute for Translational Medicine (ITM) • Maine Medical Center— U54GM115516: Northern New England Clinical & Translational Research (NNE-CTR) Network • Massachusetts General Brigham-UL1TR002541: Harvard Catalyst • Mayo Clinic Rochester-UL1TR002377: Mayo Clinic Center for Clinical and Translational Science (CCaTS) • Medical University of South Carolina—UL1TR001450: South Carolina Clinical & Translational Research Institute (SCTR) • Montefiore Medical Center-UL1TR002556: Institute for Clinical and Translational Research at Einstein and Montefiore • Nemours-U54GM104941: Delaware CTR ACCEL Program • NorthShore University HealthSystem— UL1TR002389: The Institute for Translational Medicine (ITM) • Northwestern University at Chicago—UL1TR001422: Northwestern University Clinical and Translational Science Institute (NUCATS) • OCHIN-INV-018455: Bill and Melinda Gates Foundation grant to Sage Bionetworks • Oregon Health & Science University—UL1TR002369: Oregon Clinical and Translational Research Institute • Penn State Health Milton S. Hershey Medical Center— UL1TR002014: Penn State Clinical and Translational Science Institute • Rush University Medical Center—UL1TR002389: The Institute for Translational Medicine (ITM) • Rutgers, The State University of New Jersey-UL1TR003017: New Jersey Alliance for Clinical and Translational Science • Stony Brook University-U24TR002306 • The Ohio State University-UL1TR002733: Center for Clinical and Translational Science • The State University of New York at Buffalo-UL1TR001412: Clinical and Translational Science Institute • The University of Chicago-UL1TR002389: The Institute for Translational Medicine (ITM) • The University of Iowa-UL1TR002537: Institute for Clinical and Translational Science • The University of Miami Leonard M. Miller School of Medicine-UL1TR002736: University of Miami Clinical and Translational Science Institute • The University of Michigan at Ann ArborUL1TR002240: Michigan Institute for Clinical and Health Research • The University of Texas Health Science Center at Houston-UL1TR003167: Center for Clinical and Translational Sciences (CCTS) • The University of Texas Medical Branch at Galveston—UL1TR001439: The Institute for Translational Sciences • The University of Utah-UL1TR002538: Uhealth Center for Clinical and Translational Science • Tufts Medical Center-UL1TR002544: Tufts Clinical and Translational Science Institute • Tulane University-UL1TR003096: Center for Clinical and Translational Science • University Medical Center New Orleans-U54GM104940: Louisiana Clinical and Translational Science (LA CaTS) Center • University of Alabama at Birmingham—UL1TR003096: Center for Clinical and Translational Science • University of Arkansas for Medical Sciences-UL1TR003107: UAMS Translational Research Institute • University of Cincinnati-UL1TR001425: Center for Clinical and Translational Science and Training • University of Colorado Denver, Anschutz Medical Campus-UL1TR002535: Colorado Clinical and Translational Sciences Institute • University of Illinois at Chicago–UL1TR002003: UIC Center for Clinical and Translational Science • University of Kansas Medical Center— UL1TR002366: Frontiers: University of Kansas Clinical and Translational Science Institute • University of Kentucky—UL1TR001998: UK Center for Clinical and Translational Science • University of Massachusetts Medical School Worcester-UL1TR001453: The UMass Center for Clinical and Translational Science (UMCCTS) • University of Minnesota—UL1TR002494: Clinical and Translational Science Institute • University of Mississippi Medical Center— U54GM115428: Mississippi Center for Clinical and Translational Research (CCTR) • University of Nebraska Medical Center-U54GM115458: Great Plains IDeA-Clinical & Translational Research • University of North Carolina at Chapel Hill-UL1TR002489: North Carolina Translational and Clinical Science Institute • University of Oklahoma Health Sciences Center -U54GM104938: Oklahoma Clinical and Translational Science Institute (OCTSI) • University of Rochester—UL1TR002001: UR Clinical & Translational Science Institute • University of Southern California-UL1TR001855: The Southern California Clinical and Translational Science Institute (SC CTSI) • University of Vermont–U54GM115516: Northern New England Clinical & Translational Research (NNE-CTR) Network • University of Virginia-UL1TR003015: iTHRIV Integrated Translational health Research Institute of Virginia • University of Washington—UL1TR002319: Institute of Translational Health Sciences • University of Wisconsin-Madison—UL1TR002373: UW Institute for Clinical and Translational Research • Vanderbilt University Medical Center-UL1TR002243: Vanderbilt Institute for Clinical and Translational Research • Virginia Commonwealth University—UL1TR002649: C. Kenneth and Dianne Wright Center for Clinical and Translational Research • Wake Forest University Health Sciences—UL1TR001420: Wake Forest Clinical and Translational Science Institute • Washington University in St. Louis—UL1TR002345: Institute of Clinical and Translational Sciences • Weill Medical College of Cornell University-UL1TR002384: Weill Cornell Medicine Clinical and Translational Science Center • West Virginia University—U54GM104942: West Virginia Clinical and Translational Science Institute (WVCTSI)

Submitted: Icahn School of Medicine at Mount Sinai—UL1TR001433: ConduITS Institute for Translational Sciences • The University of Texas Health Science Center at Tyler— UL1TR003167: Center for Clinical and Translational Sciences (CCTS) • University of California, Davis—UL1TR001860: UCDavis Health Clinical and Translational Science Center • University of California, Irvine—UL1TR001414: The UC Irvine Institute for Clinical and Translational Science (ICTS) • University of California, Los Angeles—UL1TR001881: UCLA Clinical Translational Science Institute • University of California, San Diego—UL1TR001442: Altman Clinical and Translational Research Institute • University of California, San Francisco —UL1TR001872: UCSF Clinical and Translational Science Institute

Pending: Arkansas Children's Hospital—UL1TR003107: UAMS Translational Research Institute • Baylor College of Medicine-None (Voluntary) • Children's Hospital of Philadelphia—UL1TR001878: Institute for Translational Medicine and Therapeutics • Cincinnati Children's Hospital Medical Center-UL1TR001425: Center for Clinical and Translational Science and Training • Emory University—UL1TR002378: Georgia Clinical and Translational Science Alliance • HonorHealth—None (Voluntary) • Loyola University Chicago— UL1TR002389: The Institute for Translational Medicine (ITM) • Medical College of Wisconsin—UL1TR001436: Clinical and Translational Science Institute of Southeast Wisconsin • MedStar Health Research Institute-UL1TR001409: The Georgetown-Howard Universities Center for Clinical and Translational Science (GHUCCTS) • MetroHealth-None (Voluntary) Montana State University—U54GM115371: American Indian/Alaska Native CTR • NYU Langone Medical Center—UL1TR001445: Langone Health's Clinical and Translational Science Institute • Ochsner Medical Center-U54GM104940: Louisiana Clinical and Translational Science (LA CaTS) Center • Regenstrief Institute–UL1TR002529: Indiana Clinical and Translational Science Institute • Sanford Research—None (Voluntary) • Stanford University -UL1TR003142: Spectrum: The Stanford Center for Clinical and Translational Research and Education • The Rockefeller University-UL1TR001866: Center for Clinical and Translational Science • The Scripps Research Institute-UL1TR002550: Scripps Research Translational Institute • University of Florida-UL1TR001427: UF Clinical and Translational Science Institute • University of New Mexico Health Sciences Center—UL1TR001449: University of New Mexico Clinical and Translational Science Center • University of Texas Health Science Center at San Antonio—UL1TR002645: Institute for Integration of Medicine and Science • Yale New Haven Hospital-UL1TR001863: Yale Center for Clinical Investigation

The content is solely the responsibility of the authors and does not necessarily represent the official views of the National Institutes of Health or N3C.

# **Author Contributions**

**Conceptualization:** Abhishek Bhatia, Alexander J. Preiss, M. Daniel Brannock, Robert F. Chew, Megan Fitzgerald, Kenneth J. Wilkins, Richard Moffitt, Emily R. Pfaff.

Formal analysis: Abhishek Bhatia, Alexander J. Preiss, Xuya Xiao, M. Daniel Brannock.

Funding acquisition: Christopher G. Chute, Melissa Haendel.

 Methodology: Abhishek Bhatia, Alexander J. Preiss, M. Daniel Brannock, Robert F. Chew, Hannah Davis, Megan Fitzgerald, Hemalkumar B. Mehta, Charisse Madlock-Brown, Kenneth J. Wilkins, Richard Moffitt, Emily R. Pfaff.

Project administration: Abhishek Bhatia, Alexander J. Preiss, Elizabeth P. Kelly.

Supervision: Abhishek Bhatia, Alexander J. Preiss, G. Caleb Alexander, Melissa Haendel, Richard Moffitt, Emily R. Pfaff.

Visualization: Abhishek Bhatia, Alexander J. Preiss, M. Daniel Brannock.

Writing – original draft: Abhishek Bhatia, Alexander J. Preiss, Xuya Xiao, M. Daniel Brannock, Megan Fitzgerald, Emily R. Pfaff.

Writing – review & editing: Abhishek Bhatia, Alexander J. Preiss, Xuya Xiao, M. Daniel Brannock, G. Caleb Alexander, Robert F. Chew, Hannah Davis, Megan Fitzgerald, Elaine Hill, Elizabeth P. Kelly, Hemalkumar B. Mehta, Charisse Madlock-Brown, Kenneth J. Wilkins, Christopher G. Chute, Melissa Haendel, Richard Moffitt, Emily R. Pfaff.

#### References

- 1. WHO Coronavirus (COVID-19) Dashboard. Accessed 2023 Mar 26. Available from: https://covid19. who.int/.
- 2. Center for Drug Evaluation, Research. FDA Updates on Paxlovid for Health Care Providers. In: U.S. Food and Drug Administration [Internet]. FDA; [accessed 2023 Mar 26]. Available from: https://www.fda.gov/drugs/news-events-human-drugs/fda-updates-paxlovid-health-care-providers.
- Hammond J, Leister-Tebbe H, Gardner A, Abreu P, Bao W, Wisemandle W, et al. Oral Nirmatrelvir for High-Risk, Nonhospitalized Adults with Covid-19. N Engl J Med. 2022; 386:1397–1408. https://doi.org/ 10.1056/NEJMoa2118542 PMID: 35172054
- Malden DE, Hong V, Lewin BJ, Ackerson BK, Lipsitch M, Lewnard JA, et al. Hospitalization and Emergency Department Encounters for COVID-19 After Paxlovid Treatment—California. 2022 Jun. p. 830– 833. https://doi.org/10.15585/mmwr.mm7125
- Yip TC-F, Lui GC-Y, Lai MS-M, Wong VW-S, Tse Y-K, Ma BH-M, et al. Impact of the Use of Oral Antiviral Agents on the Risk of Hospitalization in Community Coronavirus Disease 2019 Patients (COVID-19). Clin Infect Dis. 2023; 76:e26–e33. https://doi.org/10.1093/cid/ciac687 PMID: 36031408
- Najjar-Debbiny R, Gronich N, Weber G, Khoury J, Amar M, Stein N, et al. Effectiveness of Paxlovid in Reducing Severe Coronavirus Disease 2019 and Mortality in High-Risk Patients. Clin Infect Dis. 2023; 76:e342–e349. https://doi.org/10.1093/cid/ciac443 PMID: 35653428
- Arbel R, Wolff Sagy Y, Hoshen M, Battat E, Lavie G, Sergienko R, et al. Nirmatrelvir Use and Severe Covid-19 Outcomes during the Omicron Surge. N Engl J Med. 2022; 387:790–798. <u>https://doi.org/10. 1056/NEJMoa2204919 PMID: 36001529</u>
- Skipworth W. Covid Drug Paxlovid Now Less Effective Than In Early Trials—But It's Still Great At Preventing Death. Forbes Magazine. 21 Sep 2023. Available from: https://www.forbes.com/sites/ willskipworth/2023/09/21/covid-drug-paxlovid-now-less-effective-than-in-early-trials-but-its-still-greatat-preventing-death/. Accessed 2024 Apr 26.
- Lin D-Y, Abi Fadel F, Huang S, Milinovich AT, Sacha GL, Bartley P, et al. Nirmatrelvir or Molnupiravir Use and Severe Outcomes From Omicron Infections. JAMA Netw Open. 2023; 6:e2335077. <u>https://doi.org/10.1001/jamanetworkopen.2023.35077</u> PMID: 37733342
- Dryden-Peterson S, Kim A, Kim AY, Caniglia EC, Lennes IT, Patel R, et al. Nirmatrelvir Plus Ritonavir for Early COVID-19 in a Large U.S. Health System: A Population-Based Cohort Study. Ann Intern Med. 2023; 176:77–84. https://doi.org/10.7326/M22-2141 PMID: 36508742
- Xie Y, Bowe B, Al-Aly Z. Nirmatrelvir and risk of hospital admission or death in adults with covid-19: emulation of a randomized target trial using electronic health records. BMJ. 2023; 381:e073312. https:// doi.org/10.1136/bmj-2022-073312 PMID: 37041016
- 12. Anschuetz N. DU Study Finds History of TBI Likely Worsens Long COVID Symptoms. In: University of Denver [Internet]. 2023 Mar 28 [Accessed 2023 Dec 15]. Available from: https://www.du.edu/news/du-study-finds-history-tbi-likely-worsens-long-covid-symptoms.
- Schneider ALC, Peltz CB, Li Y, Bahorik A, Gardner RC, Yaffe K. Traumatic Brain Injury and Long-Term Risk of Stroke Among US Military Veterans. Stroke. 2023; 54:2059–2068. <u>https://doi.org/10.1161/</u> STROKEAHA.123.042360 PMID: 37334708
- 14. Granados G, Sáez-López M, Aljama C, Sampol J, Cruz M-J, Ferrer J, et al. Asbestos Exposure and Severity of COVID-19. Int J Environ Res Public Health. 2022:19. https://doi.org/10.3390/ ijerph192316305 PMID: 36498378
- Munz A. Why should veterans diagnosed with mesothelioma seek legal help? Mesothelioma Center. Vital Services for Cancer Patients & Families; 2022. Available from: <u>https://www.asbestos.com/</u>veterans/.
- Morgan RO, Teal CR, Reddy SG, Ford ME, Ashton CM. Measurement in Veterans Affairs Health Services Research: veterans as a special population. Health Serv Res. 2005; 40:1573–1583. https://doi.org/10.1111/j.1475-6773.2005.00448.x PMID: 16178996
- Klein EJ, Hardesty A, Vieira K, Farmakiotis D. Use of anti-spike monoclonal antibodies in kidney transplant recipients with COVID-19: Efficacy, ethnic and racial disparities. Am J Transplant. 2022; 22:640– 645. https://doi.org/10.1111/ajt.16843 PMID: 34591350
- Wiltz JL, Feehan AK, Molinari NM, Ladva CN, Truman BI, Hall J, et al. Racial and Ethnic Disparities in Receipt of Medications for Treatment of COVID-19—United States, March 2020-August 2021. MMWR Morb Mortal Wkly Rep. 2022; 71:96–102. https://doi.org/10.15585/mmwr.mm7103e1 PMID: 35051133
- Wu E-L, Kumar RN, Moore WJ, Hall GT, Vysniauskaite I, Kim K-YA, et al. Disparities in COVID-19 Monoclonal Antibody Delivery: a Retrospective Cohort Study. J Gen Intern Med. 2022; 37:2505–2513. https://doi.org/10.1007/s11606-022-07603-4 PMID: 35469360

- Boehmer TK, Koumans EH, Skillen EL, Kappelman MD, Carton TW, Patel A, et al. Racial and Ethnic Disparities in Outpatient Treatment of COVID-19—United States, January-July 2022. MMWR Morb Mortal Wkly Rep. 2022; 71:1359–1365. https://doi.org/10.15585/mmwr.mm7143a2 PMID: 36301738
- Hernán MA, Robins JM. Using Big Data to Emulate a Target Trial When a Randomized Trial Is Not Available. Am J Epidemiol. 2016; 183:758–764. https://doi.org/10.1093/aje/kwv254 PMID: 26994063
- Hernán MA, Wang W, Leaf DE. Target Trial Emulation: A Framework for Causal Inference From Observational Data. JAMA. 2022; 328:2446–2447. <u>https://doi.org/10.1001/jama.2022.21383</u> PMID: 36508210
- Haendel MA, Chute CG, Bennett TD, Eichmann DA, Guinney J, Kibbe WA, et al. The National COVID Cohort Collaborative (N3C): Rationale, design, infrastructure, and deployment. J Am Med Inform Assoc. 2021; 28:427–443. https://doi.org/10.1093/jamia/ocaa196 PMID: 32805036
- 24. Pfaff ER, Girvin AT, Gabriel DL, Kostka K, Morris M, Palchuk MB, et al. Synergies between centralized and federated approaches to data quality: a report from the national COVID cohort collaborative. J Am Med Inform Assoc. 2022; 29:609–618. https://doi.org/10.1093/jamia/ocab217 PMID: 34590684
- 25. Phenotype\_Data\_Acquisition: The repository for code and documentation produced by the N3C Phenotype and Data Acquisition workstream. Github. Available from: https://github.com/National-COVID-Cohort-Collaborative/Phenotype\_Data\_Acquisition.
- 26. British Medical Journal Publishing Group. Strengthening the reporting of observational studies in epidemiology (STROBE) statement: guidelines for reporting observational studies. BMJ. 2007:335. <u>https://</u> doi.org/10.1136/bmj.39386.490150.94
- 27. CDC. Interim Clinical Considerations for COVID-19 Treatment in Outpatients. In: Centers for Disease Control and Prevention [Internet]. 2023 Feb 10 [Accessed 2023 Mar 26]. Available from: https://www.cdc.gov/coronavirus/2019-ncov/hcp/clinical-care/outpatient-treatment-overview.html.
- Howard-Jones AR, Burgner DP, Crawford NW, Goeman E, Gray PE, Hsu P, et al. COVID-19 in children. II: Pathogenesis, disease spectrum and management. J Paediatr Child Health. 2022; 58:46–53. https://doi.org/10.1111/jpc.15811 PMID: 34694037
- Bose-Brill S, Hirabayashi K, Pajor NM, Rao S, Mejias A, Jhaveri R, et al. Pediatric Nirmatrelvir/Ritonavir Prescribing Patterns During the COVID-19 Pandemic. medRxiv. 2022. https://doi.org/10.1101/2022.12. 23.22283868 PMID: 36597537
- Paxlovid Drug-Drug Interactions. In: COVID-19 Treatment Guidelines [Internet]. [Accessed 2024 May 7]. Available from: https://www.covid19treatmentguidelines.nih.gov/therapies/antivirals-includingantibody-products/ritonavir-boosted-nirmatrelvir—paxlovid-/paxlovid-drug-drug-interactions/.
- **31.** Hernán MA, Sauer BC, Hernández-Díaz S, Platt R, Shrier I. Specifying a target trial prevents immortal time bias and other self-inflicted injuries in observational analyses. J Clin Epidemiol. 2016; 79:70–75. https://doi.org/10.1016/j.jclinepi.2016.04.014 PMID: 27237061
- Hripcsak G, Duke JD, Shah NH, Reich CG, Huser V, Schuemie MJ, et al. Observational Health Data Sciences and Informatics (OHDSI): Opportunities for Observational Researchers. Stud Health Technol Inform. 2015; 216:574–578. https://doi.org/10.1038/psp.2013.52 PMID: 26262116
- SPH and Sharecare Release Community Well-Being Rankings. [Accessed 2023 Mar 30]. Available from: https://www.bu.edu/sph/news/articles/2020/sph-and-sharecare-release-community-well-beingrankings/.
- 34. Hernán MA. How to estimate the effect of treatment duration on survival outcomes using observational data. BMJ. 2018; 360:k182. https://doi.org/10.1136/bmj.k182 PMID: 29419381
- Orellana L, Rotnitzky A, Robins JM. Dynamic regime marginal structural mean models for estimation of optimal dynamic treatment regimes, Part I: main content. Int J Biostat. 2010;6:Article 8. Available from: https://www.ncbi.nlm.nih.gov/pubmed/21969994. PMID: 21969994
- Hernán MA, Hernández-Díaz S, Robins JM. A structural approach to selection bias. Epidemiology. 2004; 15:615–625. https://doi.org/10.1097/01.ede.0000135174.63482.43 PMID: 15308962
- Mansournia MA, Altman DG. Inverse probability weighting. BMJ. 2016; 352:i189. <u>https://doi.org/10.1136/bmj.i189 PMID: 26773001</u>
- Willems S, Schat A, van Noorden MS, Fiocco M. Correcting for dependent censoring in routine outcome monitoring data by applying the inverse probability censoring weighted estimator. Stat Methods Med Res. 2018; 27:323–335. https://doi.org/10.1177/0962280216628900 PMID: 26988930
- Aalen OO, Cook RJ, Røysland K. Does Cox analysis of a randomized survival study yield a causal treatment effect? Lifetime Data Anal. 2015; 21:579–593. <u>https://doi.org/10.1007/s10985-015-9335-y</u> PMID: 26100005
- 40. Hernán MA, Alonso A, Logan R, Grodstein F, Michels KB, Willett WC, et al. Observational studies analyzed like randomized experiments: an application to postmenopausal hormone therapy and coronary

heart disease. Epidemiology. 2008; 19:766–779. https://doi.org/10.1097/EDE.0b013e3181875e61 PMID: 18854702

- Mendes D, Alves C, Batel-Marques F. Number needed to treat (NNT) in clinical literature: an appraisal. BMC Med. 2017; 15:112. https://doi.org/10.1186/s12916-017-0875-8 PMID: 28571585
- 42. Evaluation of Protease Inhibition for COVID-19 in Standard-Risk Patients (EPIC-SR). [Accessed 2023 Mar 31]. Available from: https://clinicaltrials.gov/ct2/show/NCT05011513.
- Jennifer H, Fountaine RJ, Carla Y, Dona F, Mary A, Weihang B, et al. Nirmatrelvir for Vaccinated or Unvaccinated Adult Outpatients with Covid-19. N Engl J Med. 2024; 390:1186–1195. https://doi.org/10. 1056/NEJMoa2309003 PMID: 38598573
- 44. CDC. Data Definitions for COVID-19 Vaccinations in the United States. In: Centers for Disease Control and Prevention [Internet]. 2023 Feb 16 [Accessed 2023 Feb 20]. Available from: https://www.cdc.gov/ coronavirus/2019-ncov/vaccines/reporting-vaccinations.html?CDC\_AA\_refVal=https%3A%2F% 2Fwww.cdc.gov%2Fcoronavirus%2F2019-ncov%2Fvaccines%2Fdistributing%2Fabout-vaccine-data. html.
- Brannock MD, Chew RF, Preiss AJ, Hadley EC, McMurry JA, Leese PJ, et al. Long COVID Risk and Pre-COVID Vaccination: An EHR-Based Cohort Study from the RECOVER Program. medRxiv. 2022. https://doi.org/10.1101/2022.10.06.22280795 PMID: 36238713
- 46. Park S, Ortega AN, Chen J, Bustamante AV. Effects of Medicare eligibility and enrollment at age 65 among immigrants and US-born residents. J Am Geriatr Soc. 2023; 71:2845–2854. https://doi.org/10. 1111/jgs.18380 PMID: 37073412
- Creators N3C Publication Committee N3C Governance Committee. N3C and Tenant Pilot Attribution and Publication Principles. https://doi.org/10.5281/zenodo.10698093
- Charlson ME, Pompei P, Ales KL, MacKenzie CR. A new method of classifying prognostic comorbidity in longitudinal studies: development and validation. J Chronic Dis. 1987; 40:373–383. <u>https://doi.org/10. 1016/0021-9681(87)90171-8 PMID: 3558716</u>
- 49. Quan H, Li B, Couris CM, Fushimi K, Graham P, Hider P, et al. Updating and validating the Charlson comorbidity index and score for risk adjustment in hospital discharge abstracts using data from 6 countries. Am J Epidemiol. 2011; 173:676–682. https://doi.org/10.1093/aje/kwq433 PMID: 21330339
- Chin T, Kahn R, Li R, Chen JT, Krieger N, Buckee CO, et al. US-county level variation in intersecting individual, household and community characteristics relevant to COVID-19 and planning an equitable response: a cross-sectional analysis. BMJ Open. 2020; 10:e039886. https://doi.org/10.1136/bmjopen-2020-039886 PMID: 32873684
- 51. Garg S, Kim L, Whitaker M, O'Halloran A, Cummings C, Holstein R, et al. Hospitalization Rates and Characteristics of Patients Hospitalized with Laboratory-Confirmed Coronavirus Disease 2019— COVID-NET, 14 States, March 1–30, 2020. MMWR Morb Mortal Wkly Rep. 2020; 69:458–464. <u>https://doi.org/10.15585/mmwr.mm6915e3 PMID</u>: 32298251
- 52. van Dorn A, Cooney RE, Sabin ML. COVID-19 exacerbating inequalities in the US. Lancet. 2020; 395:1243–1244. https://doi.org/10.1016/S0140-6736(20)30893-X PMID: 32305087
- **53.** Raifman MA, Raifman JR. Disparities in the Population at Risk of Severe Illness From COVID-19 by Race/Ethnicity and Income. Am J Prev Med. 2020; 59:137–139. https://doi.org/10.1016/j.amepre.2020.04.003 PMID: 32430225
- 54. Dyer O. Covid-19: Black people and other minorities are hardest hit in US. BMJ. 2020; 369:m1483. https://doi.org/10.1136/bmj.m1483 PMID: 32291262
- Webb Hooper M, Nápoles AM, Pérez-Stable EJ. COVID-19 and Racial/Ethnic Disparities. JAMA. 2020; 323:2466–2467. https://doi.org/10.1001/jama.2020.8598 PMID: 32391864
- Gupta S, Wang W, Hayek SS, Chan L, Mathews KS, Melamed ML, et al. Association Between Early Treatment With Tocilizumab and Mortality Among Critically III Patients With COVID-19. JAMA Intern Med. 2021; 181:41–51. https://doi.org/10.1001/jamainternmed.2020.6252 PMID: 33080002
- 57. Petito LC, García-Albéniz X, Logan RW, Howlader N, Mariotto AB, Dahabreh IJ, et al. Estimates of Overall Survival in Patients With Cancer Receiving Different Treatment Regimens: Emulating Hypothetical Target Trials in the Surveillance, Epidemiology, and End Results (SEER)-Medicare Linked Database. JAMA Netw Open. 2020; 3:e200452. https://doi.org/10.1001/jamanetworkopen.2020.0452 PMID: 32134464
- Dickerman BA, García-Albéniz X, Logan RW, Denaxas S, Hernán MA. Avoidable flaws in observational analyses: an application to statins and cancer. Nat Med. 2019; 25:1601–1606. <u>https://doi.org/10.1038/</u> s41591-019-0597-x PMID: 31591592
- Admon AJ, Donnelly JP, Casey JD, Janz DR, Russell DW, Joffe AM, et al. Emulating a Novel Clinical Trial Using Existing Observational Data. Predicting Results of the PreVent Study. Ann Am Thorac Soc. 2019; 16:998–1007. https://doi.org/10.1513/AnnalsATS.201903-241OC PMID: 31038996

- van Walraven C, Bennett C, Forster AJ. Administrative database research infrequently used validated diagnostic or procedural codes. J Clin Epidemiol. 2011; 64:1054–1059. https://doi.org/10.1016/j. jclinepi.2011.01.001 PMID: 21474278
- Wells BJ, Chagin KM, Nowacki AS, Kattan MW. Strategies for handling missing data in electronic health record derived data. EGEMS (Wash DC). 2013; 1:1035. https://doi.org/10.13063/2327-9214.1035 PMID: 25848578
- Lin KJ, Glynn RJ, Singer DE, Murphy SN, Lii J, Schneeweiss S. Out-of-system Care and Recording of Patient Characteristics Critical for Comparative Effectiveness Research. Epidemiology. 2018; 29:356– 363. https://doi.org/10.1097/EDE.00000000000794 PMID: 29283893
- 63. Wang LE, Shaw PA, Mathelier HM, Kimmel SE, French B, Evaluating Risk-Prediction Models Using Data from Electronic Health Records. Ann Appl Stat. 2016; 10:286–304. https://doi.org/10.1214/15-AOAS891 PMID: 27158296